# On the Stability of Static Poisson Networks Under Random Access

Yi Zhong, Member, IEEE, Martin Haenggi, Fellow, IEEE, Tony Q. S. Quek, Senior Member, IEEE, and Wenyi Zhang, Senior Member, IEEE

Abstract—We investigate the stable packet arrival rate region of a discrete-time slotted random access network, where the sources are distributed as a Poisson point process. Each of the sources in the network has a destination at a given distance and a buffer of infinite capacity. The network is assumed to be random but static, i.e., the sources and the destinations are placed randomly and remain static during all the time slots. We employ tools from queueing theory as well as point process theory to study the stability of this system using the concept of dominance. The problem is an instance of the interacting queues problem, further complicated by the Poisson spatial distribution. We obtain sufficient conditions and necessary conditions for stability. Numerical results show that the gap between the sufficient conditions and the necessary conditions is small when the access probability, the density of transmitters, or the SINR threshold is small. The results also reveal that a slight change of the arrival rate may greatly affect the fraction of unstable queues in the network.

*Index Terms*—Interacting queues, Poisson bipolar model, random access, stability, stochastic geometry.

## I. INTRODUCTION

A. Motivation

**T**N LARGE scale wireless networks, concurrent transmissions lead to interference between terminals. The randomness in the deployment of the transmitters makes accurate

Manuscript received November 25, 2015; revised May 2, 2016; accepted May 30, 2016. Date of publication June 7, 2016; date of current version July 12, 2016. The work of Y. Zhong and W. Zhang was supported in part by the National Basic Research Program of China (973 Program) under Grant 2012CB316004, National Natural Science Foundation of China under Grant 61379003, and the Fundamental Research Funds for the Central Universities under Grant WK350000003. The work of M. Haenggi was supported in part by the U.S. National Science Foundation (grants CCF 1216407 and CCF 1525904). The work of T. Q. S. Quek was supported in part by the A\*STAR SERC under Grant 1224104048, the MOE ARF Tier 2 under Grant MOE2014-T2-2-002, and the MOE ARF Tier 2 under Grant MOE2015-T2-2-104. This work was presented at the 2015 IEEE International Symposium on Information Theory (ISIT) [1]. The associate editor coordinating the review of this paper and approving it for publication was Z. Dawy.

Y. Zhong was with the Department of Electronic Engineering and Information Science, University of Science and Technology of China, Hefei 230027, China. He is now with the Information Systems Technology and Design Pillar, Singapore University of Technology and Design, Singapore 487372 (e-mail: geners@mail.ustc.edu.cn).

M. Haenggi is with the Department of Electrical Engineering, University of Notre Dame, Notre Dame, IN 46556 USA (e-mail: mhaenggi@nd.edu).

T. Q. S. Quek is with the Information Systems Technology and Design Pillar, Singapore University of Technology and Design, Singapore 487372 (e-mail: tonyquek@sutd.edu.sg).

W. Zhang is with the Key Laboratory of Wireless-Optical Communications, Chinese Academy of Sciences, Hefei 230027, China, and also with the Department of Electronic Engineering and Information Science, University of Science and Technology of China, Hefei 230027, China (e-mail: wenyizha@ustc.edu.cn).

Color versions of one or more of the figures in this paper are available online at http://ieeexplore.ieee.org.

Digital Object Identifier 10.1109/TCOMM.2016.2577678

modeling and analysis of the interference complicated. Recently, the introduction of the point process theory has provided great convenience for modeling and analyzing the performance of wireless networks [2]-[4]. However, most of the analytical works assume that the terminals are backlogged, i.e., that the terminals always have packets to transmit. In case where each terminal provides a buffer for queueing, the problem becomes more practically relevant and more challenging. For example, a primary problem is to study the stability of the queues in the large scale network. It can be observed from the above description that there are two issues of interest: (a) the random arrival of the packets at the terminals; (b) the noise, the fading, the interference, and the random access protocol that affect the transmission of these packets. The situation is complicated because it involves interacting queues, i.e., the serving rate of each queue depends on the sizes of all the queues. Most of previous works treat these two issues separately. The approaches based on queueing theory focus on the random arrival of the packets but ignore the physical layer as well as the effect of noise and interference [5]–[9]. Other approaches based on the multi-access information theory focus on the physical layer and analyze the transmission process but ignore the random arrival of packets [10]. The approaches based on queueing theory are often used to analyze the performance of scheduling algorithms, whereas the approaches based on the multi-access information theory mostly employ the assumption that all terminals are backlogged, and thus the results obtained constitute upper or lower bounds for the performance of certain schemes. The analysis of interacting queues requires the combination of queueing theory and multiaccess information theory and is notoriously difficult.

The analyses of interacting queues have been mostly based on the slotted ALOHA protocol with a simplified physical layer [11]. In most of the works, a discrete-time slotted ALOHA system with N terminals is considered. Each terminal maintains a buffer of infinite capacity to store the incoming packets. The time is divided into discrete slots with equal duration, and in each time slot, each terminal attempts to transmit its head-of-line packet with a certain probability if its buffer is not empty. A collision occurs if two or more terminals transmit simultaneously. When a collision occurs, all terminals involved in the collision retransmit the packet in the next time slot with the same access probability. For this simplified system, the exact stability region was characterized for two [5], [6] and three [7] terminals. When N > 3, only sufficient conditions and necessary conditions for stability were obtained [7], [12], [13]. In [14], the stability region of a queueing network with dependent servers, described by the definition

0090-6778 © 2016 IEEE. Personal use is permitted, but republication/redistribution requires IEEE permission. See http://www.ieee.org/publications\_standards/publications/rights/index.html for more information.

of subsets that can be activated simultaneously, is studied. In [15], the tradeoff between the capacity and the exact endto-end queueing delay of cell partitioned networks is analyzed and studied.

In practical wireless networks, the interference between transmissions cannot be accurately modeled as collisions. The interaction among the queues at the transmitters in practical wireless networks is thus more intricate than the aforementioned discrete-time ALOHA system. In this work, we model a large-scale wireless network using the Poisson point process (PPP). Combined with the signal-to-interference-and noise-ratio (SINR) model for successful reception, we explore the effect of random traffic arrival and queueing on the stability of large scale wireless networks.

## B. Contributions

We combine queueing theory and stochastic geometry to analyze the stability region of a static Poisson network, in which the transmitters and the receivers are placed randomly at the beginning and then remain static during all the time slots. Compared with high-mobility networks in which the nodes are regenerated independently in each time slot, the static Poisson network is more challenging to analyze since inherent correlations of the interference and signal levels persist among different time slots, due to the common randomness caused by the static locations of the nodes. Most of the practical networks are approximately static because the locations of the terminals cannot drastically change within a short time period, and the statistics obtained by spatially averaging over a large region in static networks are of great significance. From the ergodicity of the PPP, the ensemble averages obtained by averaging over the point process equal the spatial averages obtained by averaging over an arbitrary realization of the PPP over a large region. Intuitively, a direct impact of the static characteristic is that if a transmission fails at a previous time slot, there is an increased probability that it will also fail in the next few time slots [16]. If each transmitter maintains a buffer of infinite capacity to store the packets generated, the network becomes even more complicated to analyze because of the interacting queues problem. We introduce the notion of  $\epsilon$ -stability, which is a generalization of stability suitable for Poisson networks. By applying the concept of dominance [5], [17], we derive sufficient conditions and necessary conditions for  $\epsilon$ -stability. Numerical results illustrate the sufficient conditions and necessary conditions and reveal how they vary with system parameters.

# C. Related Work

Existing works about interacting queueing systems are mostly based on the discrete-time slotted random access systems in which the transmission fails when two or more terminals transmit in the same slot. Previous analyses have yielded only bounds to the stability regions [5]–[9]. Exact stability regions have been characterized only for cases when the number of terminals is two [5], [6] or three [7]. The stability and delay of multi-access systems with an infinite number of transmitters and with a simplified physical layer were studied in [18]. The work in [19] studied the geometric properties of the stability region of a slotted random access system. All these works considered the collision-based model, and the work in [17] investigated the exact stability region of SINR-based two-user interference channel.

Applications of point process theory to analyze the performance of wireless networks can be found in [2]-[4], [10], [21], and [22]. The method is widely adopted in the literature because it is analytically tractable and reflects the randomness in the practical deployment of wireless network [22], [23]. The works related to static Poisson networks include the analysis of the interference correlation [24], [25] and the local delay, which is defined as the number of time slots required for a node to successfully transmit a packet [26]-[29]. In this line of research, an implicit assumption is that the networks are backlogged. In practice, the packets arrive at each source randomly, and each source maintains a buffer to store the packets. The stability and delay of high-mobility networks were analyzed in [30] using a combination of queueing theory and stochastic geometry. In the high-mobility network, the queue sizes and the serving rates are decoupled; however, practical networks are mostly static at the time scale of the transmissions, and the decoupling exploited in high-mobility networks does not apply.

The remaining part of the paper is organized as follows. Section II describes the spatial distribution model, the arrival process, and the access protocol. Section III gives the definition of stability. Based on the concept of dominance and some related relaxations, Section IV and Section V establish sufficient conditions and necessary conditions for stability. Section VI analyzes provides asymptotic behaviors and numerical results. Finally, Section VII concludes the paper.

## II. SYSTEM MODEL

In order to analyze the stability of a large scale network, we adopt a simple yet general model. We consider a discrete-time slotted random access system with transmitters and receivers distributed as a Poisson bipolar network [3, Definition 5.8], i.e., we model the locations of the transmitters as a PPP  $\Phi = \{x_i\} \subset \mathbb{R}^d$  of intensity  $\lambda$ . Each transmitter is paired with a receiver at a fixed distance  $r_0$  and a random orientation. In the analysis, we will condition on  $x_0 \in \Phi$  at which the typical transmitter under consideration is located, where  $r_0 = |x_0|$  is the distance of this point to the origin at which the corresponding receiver is located (see Fig. 1). The time is divided into discrete slots with equal duration, and each transmission attempt occupies one time slot. We assume that the network is static, i.e., the locations of the transmitters and the receivers are generated once at the beginning and then kept unchanged during all time slots.

We use the Rayleigh block fading model in which the power fading coefficients remain static over each time slot, and are spatially and temporally independent with exponential distribution of mean 1. Let  $\alpha$  be the path loss exponent and  $h_{k,x}$  be the fading coefficient between transmitter x and the considered receiver located at the origin o in time slot k. All transmitters are assumed to transmit at unit power. The power of the thermal noise is set as W. We assume an SINR



Fig. 1. A snapshot of the Poisson bipolar network with random access.

threshold model: if the SINR over a link is above a threshold  $\theta$ , the link can be successfully used for information transmission at spectral efficiency  $\log_2(1 + \theta)$  bits/second/Hz.

Each transmitter has a buffer of infinite capacity to store the packets generated. Each transmitter generates packets according to a Bernoulli process with arrival rate  $\xi$  ( $0 \le \xi \le 1$ ) packets per time slot, i.e.,  $\xi$  is the probability of an arrival in any given time slot. In fact, the analysis and derivations in the following are not limited to the discrete Bernoulli arrival of packets since the time is slotted. For example, if the packets obey Poisson arrival, they will be served at the beginning of the next time slot if there is no packet waiting in the queue. The arrival processes of different transmitters are independent. In each time slot, each transmitter attempts to send its headof-line packet with probability p if its buffer is not empty. We assume that the feedback of the status of each attempt of transmission, either successful or failed, is instantaneous so that each transmitter is aware of the outcome. If a transmission attempt fails, the transmitter retransmits the packet in the next time slot with probability p; on the other hand, if a transmission attempt is successful, the transmitter removes the packet from the buffer.

For any time slot  $k \in \mathbb{N}^+$ , let  $\Phi_k$  be the set of transmitters that are transmitting in that time slot. The interference at the typical receiver located at the origin o in time slot k is

$$I_k = \sum_{x \in \Phi \setminus \{x_0\}} h_{k,x} |x|^{-\alpha} \mathbf{1}(x \in \Phi_k).$$
(1)

When the typical transmitter is active, the SINR of the typical receiver in time slot k is

$$\operatorname{SINR}_{k} = \frac{h_{k,x_{0}} r_{0}^{-\alpha}}{W + \sum_{x \in \Phi \setminus \{x_{0}\}} h_{k,x} |x|^{-\alpha} \mathbf{1}(x \in \Phi_{k})}.$$
 (2)

In the proposed network model, each transmitter maintains a queue with Bernoulli arrival. However, since the realization of the PPP is irregular, the distances to the interferers are different from the perspectives of individual receivers. Therefore, there are always some transmitters that experience poor performance (i.e., low success probability) while some others that experience good performance (i.e., high success probability). In view of this, even with the same arrival rate for all transmitters in a large scale network, the queues of the transmitters experiencing poor performance may become unstable because of the low success probability. Therefore, the characterization of the stability region of such networks is important and challenging.

Since we condition on  $\Phi$  having a point at  $x_0$ , the relevant probability measure of the point process is the Palm probability  $\mathbb{P}^{x_0}$ . Correspondingly, the expectation, denoted by  $\mathbb{E}^{x_0}$ , is taken with respect to the measure  $\mathbb{P}^{x_0}$ . Whether the transmission of the typical transmitter  $x_0$  is successful or not is uncertain, and the randomness comes from four aspects: the realization of PPP, the random access, the fading and the random arrival of traffic. Let  $\mathcal{C}_{\Phi}^{k}$  be the event that the transmission of the typical transmitter  $x_0$  succeeds in time slot k conditioned on the PPP  $\Phi$ .  $C_{\Phi}^{k}$  is the intersection of two events: that the transmission is scheduled by the random access scheme and that the scheduled transmission is successful. Let  $\mathbb{P}^{x_0}(\mathcal{C}^k_{\Phi}) = \mathbb{P}(\text{SINR}_k > \theta \mid \Phi, x_0 \in \Phi)$  be the success probability of the transmission of the typical transmitter  $x_0$  in time slot k conditioned on the PPP  $\Phi$ .  $\mathbb{P}^{x_0}(\mathcal{C}^k_{\Phi})$  varies with the index k because the empty or non-empty status of the queues at the interferers change over time, resulting in interference variation. In the following discussions, we will show how the stability depends on the statistical properties of  $\mathbb{P}^{x_0}(\mathcal{C}^k_{\Phi})$ .

#### III. NOTION OF $\varepsilon$ -Stability

For an isolated transmitter, by the Loynes theorem [31], if the arrival process and the serving process are stationary, the sufficient and necessary condition for stability is that the average service rate is larger than the average arrival rate. However, strict stability (i.e., all queues are stable) for a large scale network is not achievable (except for the trivial case of  $\xi = 0$ ) since there always exist some transmitters whose queues are unstable in the static Poisson network. Thus, we introduce the notion of  $\varepsilon$ -stability as follows.

Definition 1: For any  $0 \le \varepsilon \le 1$ , the  $\varepsilon$ -stability region  $S_{\varepsilon}$  is defined as

$$\mathcal{S}_{\varepsilon} \stackrel{\Delta}{=} \left\{ \xi \in \mathbb{R}^{+} : \mathbb{P}^{x_{0}} \left\{ \lim_{K \to \infty} \frac{1}{K} \sum_{k=1}^{K} \mathbb{P}^{x_{0}}(\mathcal{C}_{\Phi}^{k}) \leq \xi \right\} \leq \varepsilon \right\}.$$
(3)

Definition 2: The supremum of the  $\varepsilon$ -stability region  $S_{\varepsilon}$ , i.e.,  $\xi_{c,\varepsilon} \triangleq \sup S_{\varepsilon}$ , is called the critical arrival rate. The network is  $\varepsilon$ -stable if and only if  $\xi \leq \xi_{c,\varepsilon}$ . Remark 1:  $\mathbb{P}^{x_0} \{ \lim_{K \to \infty} \frac{1}{K} \sum_{k=1}^{K} \mathbb{P}^{x_0}(C_{\Phi}^k) \leq \xi \}$  is the

Remark 1:  $\mathbb{P}^{x_0} \{ \lim_{K \to \infty} \frac{1}{K} \sum_{k=1}^{K} \mathbb{P}^{x_0}(\mathcal{C}_{\Phi}^k) \leq \xi \}$  is the probability that the queue at the typical transmitter is unstable. We declare the network to be  $\varepsilon$ -stable if the probability for the queue at the typical transmitter being unstable is less than a threshold  $\varepsilon$  (0 <  $\varepsilon$  < 1). The probability for the queue at the typical transmitter being unstable is obtained by averaging over the point process. Based on the ergodicity of the PPP, i.e., the ensemble averages obtained by averaging over the point process equal the spatial averages obtained by averaging an arbitrary realization of the PPP over a large region, the probability that the queue at the typical transmitter is unstable transmitter is unstable transmitter.

the network. Thus, a network is  $\varepsilon$ -stable implies that the proportion of unstable transmitters in the network is less than  $\varepsilon$ .

Deriving the  $\varepsilon$ -stability region  $\mathcal{S}_{\varepsilon}$  is equivalent to obtaining the critical arrival rate  $\xi_{c,\varepsilon}$ , which is rather difficult because of the interacting queueing problem. Therefore, in the following, we obtain  $\xi_{c,\varepsilon}^{s}$  and  $\xi_{c,\varepsilon}^{n}$  with  $\xi_{c,\varepsilon}^{s} \leq \xi_{c,\varepsilon} \leq \xi_{c,\varepsilon}^{n}$ . Then,  $\xi \leq \xi_{c,\varepsilon}^{s}$  and  $\xi \leq \xi_{c,\varepsilon}^{n}$  correspond to a sufficient condition and a necessary condition for  $\varepsilon$ -stability, respectively.

For example, consider a very simple system that consists of only the typical transmission, i.e., the interference from other transmitters in the system are ignored. The success probability for that typical transmitter is  $p \exp(-W\theta r_0^{\alpha})$ . By applying the Loynes theorem, we get the condition for stability of the queue at the typical transmitter  $\xi_0$  as

$$\xi \leq \xi_0 \triangleq p \exp\left(-W\theta r_0^{\alpha}\right). \tag{4}$$

In fact, all the sufficient conditions and necessary conditions in the following sections can be expressed in the form of  $\xi \leq \beta \xi_0$  with  $0 \leq \beta \leq 1$ , where  $\xi_0$  captures the effect of noise and random access at the typical transmission while  $\beta$  captures the effect of the interference which is affected by the random access at the interfering links.

#### **IV. SUFFICIENT CONDITIONS**

In order to derive sufficient conditions for  $\varepsilon$ -stability, we consider a dominant system [5], [9], [17]. In the dominant system the typical transmitter behaves exactly the same as in the original system. However, for the other transmitters in the dominant system, when the queue at a transmitter becomes empty, it continues to transmit "dummy" packets with the access probability p, thus continuing to cause interference to other transmissions with probability p. So in the dominant system, the queue size at each transmitter is always no smaller than that in the original system, provided the queues start with the same initial conditions. In the dominant system, the success probability given  $\Phi$  is the same for different time slots because all transmitters always have packets to transmit, and the fading and the scheduling result of random access are i.i.d. between different time slots. The  $\varepsilon$ -stability region  $S_{\varepsilon}$  is simplified into  $\overline{\mathcal{S}}_{\varepsilon} = \{ \xi \in \mathbb{R}^+ : \mathbb{P}^{x_0} \{ \mathbb{P}^{x_0}(\mathcal{C}_{\Phi}) \leq \xi \} \leq \varepsilon \}$ . By deriving the  $\varepsilon$ -stability conditions for the dominant system, we get a sufficient condition for the original system to be  $\varepsilon$ -stable.

Theorem 1: Given a slotted random access system with the transmitters distributed as a PPP and with Bernoulli packet arrivals, a sufficient condition for the system to be  $\varepsilon$ -stable is

$$\xi \le \xi_{c,\varepsilon}^{s},\tag{5}$$

where  $\zeta_{\mathbf{c},\varepsilon}^{\mathbf{s}} \triangleq \sup \overline{\mathcal{S}}_{\varepsilon}$  is given by

$$\begin{aligned} \xi_{\mathbf{c},\varepsilon}^{\mathbf{s}} &= \sup\left\{ \boldsymbol{\xi} \in \mathbb{R}^{+} : \frac{1}{2} - \frac{1}{\pi} \int_{0}^{\infty} \frac{1}{\omega} \\ &\operatorname{Im}\left\{ \left(\frac{\boldsymbol{\xi}_{0}}{\boldsymbol{\xi}}\right)^{j\omega} e^{-j\omega C_{\delta 2} F_{1}(1-j\omega, 1-\delta; 2; p)} \right\} \mathrm{d}\boldsymbol{\omega} \leq \varepsilon \right\}, \end{aligned}$$
(6)

with  $\delta = 2/\alpha$ ,  $C_{\delta} = p\lambda \pi r_0^2 \theta^{\delta} \Gamma(1 + \delta) \Gamma(1 - \delta)$ , and  $_2F_1(a, b; c; z)$  is the Gaussian hypergeometric function. Thus, a lower bound on the critical arrival rate  $\xi_{c,\varepsilon}$  is  $\xi_{c,\varepsilon}^{s}$ , *i.e.*,  $\xi_{c,\varepsilon} \geq \xi_{c,\varepsilon}^{s}$ .

*Proof:* See Appendix A. 
$$\Box$$

*Remark 2:*  $\xi_{c,\varepsilon}^s$  given by (6) can be written as

where  $\xi_0/\xi$  in (6) is replaced by  $1/\beta$ , and  $\beta$  is the parameter introduced after (4) that captures the effect of the interference. As  $\lambda \to 0$ , we have  $C_{\delta} \to 0$ , and (7) becomes

$$\xi_{\mathsf{c},\varepsilon}^{\mathsf{s}} = \xi_0 \sup \left\{ \beta \in \mathbb{R}^+ : \frac{1}{2} + \frac{1}{\pi} \int_0^\infty \frac{\sin(\omega \ln \beta)}{\omega} \mathrm{d}\omega \le \varepsilon \right\}.$$
(8)

Since  $\int_{-\infty}^{0} \frac{\sin(\pi x)}{\pi x} dx = \int_{0}^{\infty} \frac{\sin(\pi x)}{\pi x} dx = \frac{1}{2}$ , when  $\ln \beta > 0$ , the expression  $\frac{1}{2} + \frac{1}{\pi} \int_{0}^{\infty} \frac{1}{\omega} \sin(\omega \ln \beta) d\omega$  evaluates to 1; otherwise when  $\ln \beta < 0$  it evaluates to 0. Thus, we have

$$\begin{aligned} \xi_{\mathbf{c},\varepsilon}^{\mathbf{s}} &= \xi_{\mathbf{0}} \sup \left\{ \beta \in \mathbb{R}^{+} : \mathbf{1} \left( \ln \beta > 0 \right) \le \varepsilon \right\} \\ &= \xi_{\mathbf{0}}, \end{aligned} \tag{9}$$

where  $\mathbf{1}(\cdot)$  is the indicator function. This is exactly the case where the interference is ignored and only noise and fading affect the transmission. Combined with the necessary condition (4)  $\xi \leq \xi_0$ , we get the exact critical arrival rate  $\xi_{c,\varepsilon} = \xi_0 \text{ as } \lambda \to 0$ .

As  $\theta \to 0$ , the sufficient condition (7) becomes

$$\begin{aligned} \xi_{\mathbf{c},\varepsilon}^{s} &= \sup\left\{ \xi \in \mathbb{R}^{+} : \frac{1}{2} - \frac{1}{\pi} \int_{0}^{\infty} \frac{\sin\left(\omega \ln p - \omega \ln \xi\right)}{\omega} \mathrm{d}\omega \leq \varepsilon \right\} \\ &= p. \end{aligned}$$
(10)

As  $\theta \to 0$ , the necessary condition (4) becomes  $\xi \leq p$ . Thus, we get the exact critical arrival rate  $\xi_{c,\varepsilon} = p$  as  $\theta \to 0$ .

In the original system where the queues interact with each other, if  $\lambda \to 0$ , the interference is negligible, and a transmission in the network is only affected by the thermal noise. And if  $\theta \to 0$ , a transmission is almost surely successful if it is scheduled. Therefore, in these cases, the serving processes of the packets at different transmitters can be approximated as decoupled and independent, and the critical arrival rates above are intuitively reasonable.

The following corollary gives a closed-form sufficient condition that is weaker than the one given by Theorem 1 but easier to evaluate.

Corollary 1: Given a slotted random access system with the transmitters distributed as a PPP and with Bernoulli arrivals, a sufficient condition for the system to be  $\varepsilon$ -stable is

$$\xi \le \widetilde{\zeta}_{\mathbf{c},\varepsilon}^{\mathbf{s}},\tag{11}$$

where  $\widetilde{\zeta}_{c,\varepsilon}^{s} \triangleq \max_{n \in \mathbb{N}^+} \eta(n)$ , and

$$\eta(n) = \xi_0 \varepsilon^{\frac{1}{n}} \exp\left(-\pi \lambda \delta (1-p)^{\delta} \theta^{\delta} r_0^2\right)$$
$$\sum_{i=1}^n ((1-p)^{-i} - 1) \frac{\Gamma(i-\delta)\Gamma(n-i+\delta)}{\Gamma(i+1)\Gamma(n-i+1)} \right). (12)$$

Thus, a closed-form lower bound on the critical arrival rate  $\xi_{c,\varepsilon}$  is  $\tilde{\xi}_{c,\varepsilon}^{s}$ , i.e.,  $\xi_{c,\varepsilon} \geq \tilde{\xi}_{c,\varepsilon}^{s}$ . *Proof:* For all  $n \in \mathbb{N}^+$ , the cdf of  $\mathbb{P}^{x_0}(\mathcal{C}_{\Phi})$  is

$$\mathbb{P}^{x_0}\left\{\mathbb{P}^{x_0}(\mathcal{C}_{\Phi}) \leq \zeta\right\} = \mathbb{P}^{x_0}\left\{e^{-n\ln(\mathbb{P}^{x_0}(\mathcal{C}_{\Phi}))} \geq e^{-n\ln\zeta}\right\}.$$
 (13)

By applying the Markov inequality, we obtain

$$\mathbb{P}^{x_{0}} \left\{ \mathbb{P}^{x_{0}}(\mathcal{C}_{\Phi}) < \zeta \right\} < \frac{1}{e^{-n \ln \zeta}} \mathbb{E} \left[ e^{-n \ln(\mathbb{P}^{x_{0}}(\mathcal{C}_{\Phi}))} \right] \\ = p^{-n} \exp\left(n \ln \zeta + n\theta r_{0}^{a} W\right) \\ \times \mathbb{E} \left[ \prod_{x \in \Phi \setminus \{x_{0}\}} \left( \frac{p}{1 + \theta r_{0}^{a} |x|^{-a}} + 1 - p \right)^{-n} \right] \right] \\ = \left( \frac{\zeta}{\zeta_{0}} \right)^{n} \exp\left( - 2\pi \lambda \right) \\ \times \int_{0}^{\infty} \left( 1 - \left( \frac{p}{1 + \theta r_{0}^{a} r^{-a}} + 1 - p \right)^{-n} \right) r dr \right) \\ = \left( \frac{\zeta}{\zeta_{0}} \right)^{n} \exp\left( 2\pi \lambda \right) \\ \times \int_{0}^{\infty} \frac{(1 + \theta r_{0}^{a} r^{-a})^{n} - (1 + (1 - p)\theta r_{0}^{a} r^{-a})^{n}}{(1 + (1 - p)\theta r_{0}^{a} r^{-a})^{n}} r dr \right) \\ \stackrel{(a)}{=} \left( \frac{\zeta}{\zeta_{0}} \right)^{n} \exp\left( 2\pi \lambda \sum_{i=0}^{n} C_{n}^{i} (1 - (1 - p)^{i}) \right) \\ \times \int_{0}^{\infty} \frac{(\theta r_{0}^{a} r^{-a})^{i} r}{(1 + (1 - p)\theta r_{0}^{a} r^{-a})^{n}} dr \right) \\ \stackrel{(b)}{=} \left( \frac{\zeta}{\zeta_{0}} \right)^{n} \exp\left( \pi \lambda n \delta (1 - p)^{\delta} \theta^{\delta} r_{0}^{2} \right) \\ \times \sum_{i=1}^{n} ((1 - p)^{-i} - 1) \frac{\Gamma(i - \delta) \Gamma(n - i + \delta)}{\Gamma(i + 1) \Gamma(n - i + 1)} \right).$$

$$(14)$$

where  $C_n^i = n!/(i!(n-i)!) = \Gamma(n+1)/(\Gamma(i+1)\Gamma(n-i+1))$ is the binomial coefficient. (*a*) holds from the binomial expansion and the exchange of summation and integral. (*b*) follows from the relationship between the beta function  $B(x, y) = \int_0^1 t^{x-1}(1-t)^{y-1} dt$  and the gamma function and from the fact that the term for i = 0 equals zero.

Since the above inequality holds for all  $n \in \mathbb{N}^+$ , we have

$$\overline{\mathcal{S}}_{\varepsilon} \supset \bigcup_{n \in \mathbb{N}^{+}} \left\{ \xi \in \mathbb{R}^{+} : \left(\frac{\xi}{\xi_{0}}\right)^{n} \\ \times \exp\left(\pi \lambda n \delta(1-p)^{\delta} \theta^{\delta} r_{0}^{2} \sum_{i=1}^{n} ((1-p)^{-i}-1) \right) \\ \times \frac{\Gamma(i-\delta)\Gamma(n-i+\delta)}{\Gamma(i+1)\Gamma(n-i+1)} \leq \varepsilon \right\}.$$
(15)

Taking the supremum on both sides of (15) results in

$$\sup \overline{\mathcal{S}}_{\varepsilon} > \max_{n \in \mathbb{N}^+} \eta(n), \tag{16}$$

where  $\eta(n)$  is given by (12). Letting  $\widetilde{\zeta}_{c,\varepsilon}^s = \max_{\eta \in \mathbb{N}^+} \eta(n)$ , we get  $\widetilde{\zeta}_{c,\varepsilon}^s < \sup \overline{\mathcal{S}}_{\varepsilon} = \zeta_{c,\varepsilon}^s$ , indicating that  $\zeta \leq \widetilde{\zeta}_{c,\varepsilon}^s$  is also a sufficient condition for  $\varepsilon$ -stability which is "looser" than  $\zeta \leq \zeta_{c,\varepsilon}^s$ .



Fig. 2. The simplified system which consists of two pairs of transceivers, i.e., the typical transmission and the nearest interfering transmission in the original system.

## V. NECESSARY CONDITIONS

The simple condition given in (4) is weak because it ignores the interference. In the following, we propose two approaches to derive two different types of necessary conditions for  $\varepsilon$ -stability. In the derivation of the type I necessary conditions, we consider a simplified system in which only the effect of the nearest interferer is considered. Since the interference is reduced in the simplified system, a necessary condition for the typical transmitter to be stable in the original system is that it is stable in the simplified system. In the derivation of the type II necessary conditions, we consider a modified favorable system that drops the packets in the interfering transmitters that are not scheduled by the random access or whose transmission failed. Since the interference is not larger than that in the original system and the packets will not accumulate at the interfering transmitters, the  $\varepsilon$ -stability region will be a subset of that of the original system.

## A. Type I Necessary Conditions

First we derive type I necessary conditions and consider a simplified version of the original system, in which only two pairs of transmitters and receivers are considered. One pair is the typical pair in the original system, whose transmitter is located at  $x_0 = (r_0, 0)$  and the corresponding receiver  $y_0$  is located at the origin o. The other pair is the pair containing the nearest interferer. Let  $x_1 = (r_m \cos \varphi, r_m \sin \varphi)$  be the location of the nearest transmitter, where  $r_m$  is the distance from the origin and  $\varphi$  is the angle. Let  $y_1 = (r_m \cos \varphi + r_0 \cos \psi, r_m \sin \varphi + r_0 \sin \psi)$  be the location of the associated receiver, where  $\psi$  is the angle between  $x_1$  and  $y_1$  (see Fig. 2).  $\varphi$  and  $\psi$  are independent uniformly distributed random variables in  $[0, 2\pi]$ . The pdf of  $r_m$  is

$$f_{r_{\rm m}}(r) = 2\pi \,\lambda r \exp\left(-\pi \,\lambda r^2\right). \tag{17}$$

A necessary condition for the original system to be  $\varepsilon$ -stable is that the probability of the transmitter located at  $x_0$  in the simplified system being unstable is less than  $\varepsilon$ . Notice that we only need to consider the stability of the queue at the transmitter  $x_0$ , i.e., it does not matter whether the interfering transmitter's queue is stable or not. Since  $r_m$  is a random variable, it is uncertain whether the queue at the transmitter  $x_0$  is stable or not. However, if  $r_m$  is given, the stability of the queue at the transmitter  $x_0$  is determined. Therefore, we first derive a sufficient and necessary condition for the transmitter  $x_0$  to be stable when  $r_m$  is given.

Consider a dominant system of the simplified system, i.e., the transmitter  $x_0$  still transmits "dummy" packets when its queue is empty, thus it keeps causing interference. Unlike the transmitter  $x_0$ , the nearest interfering transmitter  $x_1$  in the dominant system behaves the same as in the original simplified system. In fact, a sufficient and necessary condition for the transmitter  $x_0$  in the simplified system to be stable is that it is stable in the dominant simplified system. The sufficiency claims that if the queue at  $x_0$  is stable in the dominant simplified system, then it will be stable in the original simplified system. This is because the interference in the dominant simplified system is larger than that in the original simplified system, resulting in smaller success probability. The necessity claims that if the queue at  $x_0$  is unstable in the dominant simplified system, then it will be unstable in the original simplified system. This is because when the queue at the transmitter  $x_0$  in the dominant simplified system is unstable, the queue size will grow to infinity without emptying with non-zero probability. Notice that as long as the queue at  $x_0$  is not empty, the dominant simplified system and the original simplified system behave identically if starting from the same initial condition, and the dominant simplified system is indistinguishable from the original simplified system under saturation. Thus the sample paths that do not visit queue size zero in the dominant simplified system also occur in the original simplified system, and they constitute a positive measure of all sample paths. Therefore, the queue at  $x_0$  in the original simplified system is also unstable. Combining the two parts, the proof of the sufficiency and the necessity is complete. Therefore, we only need to derive the sufficient and necessary condition for the transmitter  $x_0$  to be stable in the dominant simplified system. Based on these ideas, we get the following lemma.

Lemma 1: For the simplified system with given  $\varphi, \psi, r_m$ , the sufficient and necessary condition for the queue at the transmitter  $x_0$  to be stable is

$$\xi \leq \begin{cases} \xi_0 \frac{(1+(1-p)\theta_{\rm s})(1+\theta_{\rm m})}{p(\theta_{\rm m}-\theta_{\rm s})+(1+\theta_{\rm s})(1+\theta_{\rm m})} & \text{if } r > r_{\rm m} \\ \xi_0 \frac{1+(1-p)\theta_{\rm m}}{1+\theta_{\rm m}} & \text{if } r_{\rm s} \le r_{\rm m} \end{cases}$$
(18)

where  $\theta_{\rm s} = \theta r_0^{\alpha} r_{\rm s}^{-\alpha}$ ,  $\theta_{\rm m} = \theta r_0^{\alpha} r_{\rm m}^{-\alpha}$  and  $r_{\rm s} = \sqrt{(r_{\rm m} \cos \varphi + r_0 \cos \psi - r_0)^2 + (r_{\rm m} \sin \varphi + r_0 \sin \psi)^2}$ . *Proof:* See Appendix B.

Lemma 1 gives the sufficient and necessary condition for the queue at the transmitter  $x_0$  to be stable with given  $\varphi$ ,  $\psi$ ,  $r_m$ . For the nearest interferer,  $\varphi$ ,  $\psi$ ,  $r_m$  are random variables. By applying the results in Lemma 1, we get the following theorem.

Theorem 2: Given a slotted random access system with the transmitters distributed as a PPP and with Bernoulli packet arrivals, a type I necessary condition for the system to be



Fig. 3. The simplified system when  $\varphi = \psi = -\pi$  which consists of two pairs of transceivers.

*ɛ*-stable is

$$\xi \leq \xi_{\mathbf{c},\varepsilon}^{\mathbf{n}1} \triangleq \xi_0 \left( 1 - \frac{\theta p}{\theta + \left(F_Z^{-1}(\varepsilon)\right)^{\alpha}} \right), \tag{19}$$

where  $Z = \frac{1}{r_0} \max\{r_m, r_s\}$  with  $F_Z(z)$  being the cdf of Z, and  $r_s$  is defined in Lemma 1. Thus, a upper bound on the critical arrival rate  $\xi_{c,\varepsilon}$  is  $\xi_{c,\varepsilon}^{n1}$ , i.e.,  $\xi_{c,\varepsilon} \leq \xi_{c,\varepsilon}^{n1}$ .

*Proof:* See Appendix C.

The necessary condition given by Theorem 2 is not in closed form, and thus the necessary condition needs to be obtained through numerical evaluation. In the following, we derive a closed-form necessary condition by considering the further simplified system with  $\varphi = \psi = -\pi$  (see Fig. 3). For a given  $r_m$  if the transmitter  $x_0$  in the simplified system is unstable for  $\varphi = \psi = -\pi$ , it will also be unstable for other values of  $\varphi$  and  $\psi$ . This is because when  $\varphi = \psi = -\pi$ , the interference between the two pairs of transceivers is the smallest among all  $\varphi$  and  $\psi$ . The following lemma gives the sufficient and necessary condition for the queue at the transmitter  $x_0$  to be stable when  $\varphi = \psi = -\pi$  with given  $r_m$ .

Lemma 2: For the simplified system with  $\varphi = \psi = -\pi$  and given  $r_{\rm m}$  (see Fig. 3), the sufficient and necessary condition for the queue at the transmitter  $x_0$  to be stable is

$$\xi \leq \xi_0 \frac{(1+(1-p)\theta_{\rm s})(1+\theta_{\rm m})}{p(\theta_{\rm m}-\theta_{\rm s})+(1+\theta_{\rm s})(1+\theta_{\rm m})},\tag{20}$$

where  $\theta_{\rm s} = \theta r_0^{\alpha} r_{\rm s}^{-\alpha} = \theta r_0^{\alpha} (r_{\rm m} + 2r_0)^{-\alpha}$  and  $\theta_{\rm m} = \theta r_0^{\alpha} r_{\rm m}^{-\alpha}$ .

*Proof:* This lemma is a special case of Lemma 1 obtained by setting  $\varphi = \psi = -\pi$ .

In Lemma 2, the case where  $\varphi = \psi = -\pi$  is considered. For any other  $\varphi$  and  $\psi$  with given  $r_m$ , (20) gives a necessary condition for the queue at the transmitter  $x_0$  to be stable in the simplified system. Since  $r_m$  is a random variable and its probability distribution is given by (17), (20) gives a necessary condition for the queue at the transmitter  $x_0$  to be stable in the simplified system. The simplified system only considers the interference from the nearest transmitter; thus (20) will also be a necessary condition for  $\varepsilon$ -stability of the original system. By modifying the proof of Theorem 2 with  $r_s = r_m + 2r_0$ , we obtain the following corollary.

Corollary 2: Given a slotted random access system with the transmitters distributed as a PPP and with Bernoulli packet arrivals, a closed-form type I necessary condition for  $\varepsilon$ -stability is

$$\xi \leq \tilde{\zeta}_{c,\varepsilon}^{n1} \triangleq \xi_0 \left( 1 + \left( \frac{p \theta r_0^{\alpha}}{\left( \sqrt{-\frac{\ln(1-\varepsilon)}{\pi \lambda}} + 2r_0 \right)^{\alpha} + \theta r_0^{\alpha}} \right)^2 \right)^{-1}.$$
(21)

Thus, a closed-form upper bound on the critical arrival rate  $\xi_{c,\varepsilon}$  is  $\widetilde{\xi}_{c,\varepsilon}^{n1}$ , i.e.,  $\xi_{c,\varepsilon} \leq \widetilde{\xi}_{c,\varepsilon}^{n1}$ .

*Proof:* See Appendix D. 
$$\Box$$

# B. Type II Necessary Conditions

In the following, we derive the type II necessary conditions. We consider a modified favorable system, in which the packets in the interfering transmitters that are not scheduled by random access or whose transmission failed will be dropped instead of being retransmitted; thus, an interfering transmitter is active with probability  $p\xi$ , decoupled from the status of other transmitters.

Theorem 3: Given a slotted random access system with the transmitters distributed as a PPP and with Bernoulli packet arrivals, a type II necessary condition for the system to be  $\varepsilon$ -stable is

$$\begin{split} \xi &\leq \xi_{\mathsf{c},\varepsilon}^{\mathsf{n}2} \triangleq \sup\left\{\xi \in \mathbb{R}^{+} : \frac{1}{2} - \frac{1}{\pi} \right. \\ &\int_{0}^{\infty} \frac{1}{\omega} \mathrm{Im}\left\{\left(\frac{\xi_{0}}{\xi}\right)^{j\omega} e^{-j\omega\xi C_{\delta 2}F_{1}(1-j\omega, 1-\delta; 2; \xi p)}\right\} \mathrm{d}\omega \leq \varepsilon \right\}. \end{split}$$

Thus, a upper bound on the critical arrival rate  $\xi_{c,\varepsilon}$  is  $\xi_{c,\varepsilon}^{n2}$ *i.e.*,  $\xi_{c,\varepsilon} \leq \xi_{c,\varepsilon}^{n2}$ . *Proof:* See Appendix E.

Remark 3: If  $\lambda \to 0$  or  $\theta \to 0$ , the necessary condition in Theorem 3 coincides with the sufficient condition in Theorem 1, indicating that the two conditions are tight for small  $\lambda$  and  $\theta$ .

The following corollary simplifies the type II necessary condition using the Markov inequality.

*Corollary 3: Given a slotted random access system with the* transmitters distributed as a PPP and with Bernoulli packet arrivals, a type II necessary condition for the system to be  $\varepsilon$ -stable is

$$\xi \leq \xi_0 (1-\varepsilon)^{-\frac{1}{t}} \exp\left(-\xi C_{\delta 2} F_1(1-t, 1-\delta; 2; \xi p)\right),$$
(23)

for all t > 0. For t = 1, we obtain a closed-form type II necessary condition as

$$\xi \leq \tilde{\xi}_{c,\varepsilon}^{n2} \triangleq \frac{1}{C_{\delta}} \mathcal{W}\left(\frac{C_{\delta}\xi_{0}}{1-\varepsilon}\right), \tag{24}$$

where W(z) is the main branch of Lambert W function. Thus, a closed-form upper bound on the critical arrival rate  $\xi_{c,\varepsilon}$  is  $\widetilde{\xi}_{c,\varepsilon}^{n2}$ , *i.e.*,  $\xi_{c,\varepsilon} \leq \widetilde{\xi}_{c,\varepsilon}^{n2}$ .

Proof: See Appendix F.

When deriving the type I necessary condition, we only considered the effect of the nearest interferer and ignored all other interferers, while in the derivation of the type II necessary condition, we considered all interferers but ignored the retransmission mechanism of the interferers. Whether the type I or the type II necessary condition should be used depends on whether the nearest interferer or the retransmission mechanism of the interferers takes the leading position in affecting the transmission. For example, when the SINR threshold  $\theta$  is small and the access probability p is large, the packets will be highly likely scheduled and transmitted successfully, and no retransmission happens. Therefore, the effect of the retransmission mechanism can be neglected, and the type II necessary condition is better than the type I necessary condition.

#### VI. DISCUSSION OF RESULTS

While the conditions given by the theorems are not in closed form, the corollaries give closed-form results. In order to gain insight from the results, we discuss the asymptotic behaviors and compare the sufficient and necessary conditions through numerical evaluations.

#### A. Asymptotic Behaviors

1) p Approaching 0: From Corollary 1, as  $p \to 0$ , the optimal *n* is  $n_{\text{max}} = \infty$ . Thus, we have

$$\lim_{p \to 0} \tilde{\xi}_{\mathbf{c},\varepsilon}^{\mathbf{s}} = \xi_0.$$
<sup>(25)</sup>

From Corollary 2, we get

$$\lim_{p \to 0} \tilde{\xi}_{c,\varepsilon}^{n1} = \xi_0.$$
<sup>(26)</sup>

From Corollary 3 and by noticing that  $\lim_{z\to 0} \mathcal{W}(z)/z = 1$ , we get

$$\lim_{p \to 0} \tilde{\zeta}_{c,\varepsilon}^{n2} = \frac{\zeta_0}{1 - \varepsilon}.$$
(27)

2)  $\varepsilon$  Approaching 0: Corollary 1 shows that  $\xi_{\varepsilon,\varepsilon}^s$  approaches zero exponentially with attenuation factor  $\frac{1}{n_{\text{max}}}$  as  $\varepsilon \to 0$ . From Corollary 2, we get the asymptotic result for  $\tilde{\zeta}_{c.\epsilon}^{n1}$ as  $\varepsilon \to 0$  as

$$\begin{aligned} \tilde{\xi}_{c,\varepsilon}^{n1} &= \tilde{\xi}_0 \left( 1 + \left( \frac{p\theta r_0^a}{\left( \sqrt{-\frac{\ln(1-\varepsilon)}{\pi\lambda}} + 2r_0 \right)^a + \theta r_0^a} \right)^2 \right)^{-1} \\ &= \frac{(2^a + \theta)^2}{p^2 \theta^2 + (2^a + \theta)^2} \tilde{\xi}_0 + O\left( -\frac{\ln(1-\varepsilon)}{\pi\lambda} \right) \\ &+ \frac{a2^a p^2 \theta^2 (2^a + \theta)}{((2^a + \theta)^2 + p^2 \theta^2)^2 r_0} \tilde{\xi}_0 \sqrt{-\frac{\ln(1-\varepsilon)}{\pi\lambda}} \\ &= \frac{(2^a + \theta)^2}{p^2 \theta^2 + (2^a + \theta)^2} \tilde{\xi}_0 \\ &+ \frac{a2^a p^2 \theta^2 (2^a + \theta)}{((2^a + \theta)^2 + p^2 \theta^2)^2 r_0 \sqrt{\pi\lambda}} \tilde{\xi}_0 \varepsilon^{\frac{1}{2}} + O(\varepsilon). \end{aligned}$$
(28)

(28) shows that  $\tilde{\xi}_{c,\varepsilon}^{n1}$  approaches  $\frac{(2^{\alpha}+\theta)^2}{p^2\theta^2+(2^{\alpha}+\theta)^2}\xi_0$  with residual  $\Theta(\varepsilon^{\frac{1}{2}}).$ 

From Corollary 3, letting  $z_0 = \xi_0 C_{\delta}$ , we get the asymptotic results for  $\tilde{\xi}_{c,\varepsilon}^{n_2}$  as  $\varepsilon \to 0$  as

$$\widetilde{\xi}_{c,\varepsilon}^{n2} \stackrel{(a)}{=} \frac{\mathcal{W}(z_0)}{C_{\delta}} + \frac{\mathcal{W}(z_0)z_0}{C_{\delta}z_0(1+\mathcal{W}(z_0))} \frac{\varepsilon}{1-\varepsilon} + O\left(\frac{\varepsilon}{1-\varepsilon}\right) \\ = \frac{\mathcal{W}(z_0)}{C_{\delta}} + \frac{\mathcal{W}(z_0)}{C_{\delta}(1+\mathcal{W}(z_0))}\varepsilon + O(\varepsilon^2)$$
(29)

where (a) follows from the Taylor expansion of  $\mathcal{W}(z)$  at  $z_0$ , i.e.  $\mathcal{W}(z) = \mathcal{W}(z_0) + \frac{\mathcal{W}(z_0)(z-z_0)}{z_0(1+\mathcal{W}(z_0))} + O((z-z_0)^2)$  as  $z \to z_0$ . (29) shows that  $\tilde{\zeta}_{c,\varepsilon}^{n2}$  approaches  $\frac{\mathcal{W}(z_0)}{C_{\delta}}$  with residual  $\Theta(\varepsilon)$ . 3)  $\lambda$  Approaching 0: From Corollary 1, as  $\lambda \to 0$ , the

3)  $\lambda$  Approaching 0: From Corollary 1, as  $\lambda \to 0$ , the optimal *n* to maximize  $\eta(n)$  is  $n_{\max} = \infty$ . The asymptotic result for  $\tilde{\zeta}_{c,\varepsilon}^s$  as  $\lambda \to 0$  is

$$\widetilde{\xi}_{c,\varepsilon}^{s} = \xi_0 \bigg( 1 - \pi \,\lambda \delta (1-p)^{\delta} \theta^{\delta} r_0^2 \lim_{n \to \infty} \sum_{i=1}^n ((1-p)^{-i} - 1) \\ \times \frac{\Gamma(i-\delta)\Gamma(n-i+\delta)}{\Gamma(i+1)\Gamma(n-i+1)} + O(\lambda^2) \bigg), \tag{30}$$

which reveals that  $\tilde{\zeta}_{c,\varepsilon}^s$  approaches  $\xi_0$  with a factor of  $1 - \Theta(\lambda)$ .

From Corollary 2, we get the asymptotic result for  $\widetilde{\xi}_{c,\varepsilon}^{nl}$  as  $\lambda \to 0$  as

$$\widetilde{\xi}_{c,\varepsilon}^{n1} = \widetilde{\xi}_0 \bigg( 1 - \frac{p^2 \theta^2 r_0^{2\alpha} \pi^{\alpha}}{(-\ln(1-\varepsilon))^{\alpha}} \lambda^{\alpha} + O(\lambda^{2\alpha}) \bigg), \quad (31)$$

indicating that  $\tilde{\xi}_{c,\varepsilon}^{n1}$  approaches  $\xi_0$  with a factor of  $1 - \Theta(\lambda^{\alpha})$ . From Corollary 3 and by noticing that  $\mathcal{W}(z) = z - O(z^2)$  as  $z \to 0$ , we get the asymptotic results for  $\tilde{\xi}_{c,\varepsilon}^{n2}$  as  $\lambda \to 0$  as

$$\widetilde{\xi}_{c,\varepsilon}^{n2} = \frac{1}{1-\varepsilon} \widetilde{\xi}_{0} \\ -\frac{1}{(1-\varepsilon)^{2}} p \lambda \pi r_{0}^{2} \theta^{\delta} \Gamma(1+\delta) \Gamma(1-\delta) (\widetilde{\xi}_{0})^{2} + O(\lambda^{2}),$$
(32)

which reveals that  $\tilde{\zeta}_{c,\varepsilon}^{n2}$  approaches  $\frac{1}{1-\varepsilon}\xi_0$  with residual  $\Theta(\lambda)$ .

4)  $\theta$  Approaching  $\theta$ : When fixing the duration of each time slot and varying  $\theta$ , we multiply  $\tilde{\xi}_{c,\varepsilon}^s$  with the factor  $\log_2(1+\theta)$ since the size of each packet is changed. The factor  $\log_2(1+\theta)$ guarantees that when varying  $\theta$ , arrival rates with different packet sizes are compared fairly. From Corollary 1, as  $\theta \to 0$ , the optimal *n* to maximize  $\eta(n)$  is  $n_{\max} = \infty$ . We get the asymptotic results for  $\tilde{\xi}_{c,\varepsilon}^s \log_2(1+\theta)$  as  $\theta \to 0$  as

$$\widetilde{\xi}_{c,\varepsilon}^{s} \log_2(1+\theta) = \frac{p}{\ln 2}\theta + O(\theta^2).$$
(33)

From Corollary 2, we get the asymptotic results for  $\widetilde{\zeta}_{c.\varepsilon}^{n1} \log_2(1+\theta)$  as  $\theta \to 0$  as

$$\widetilde{\xi}_{c,\varepsilon}^{n1}\log_2(1+\theta) = \frac{p}{\ln 2}\theta + O(\theta^2).$$
(34)

From Corollary 3 and by noticing that  $\lim_{z\to 0} W(z)/z = 1$ , we get the asymptotic results for  $\tilde{\xi}_{c,\varepsilon}^{n2} \log_2(1 + \theta)$  as  $\theta \to 0$  as

$$\widetilde{\xi}_{c,\varepsilon}^{n2} \log_2(1+\theta) = \frac{p\theta}{(1-\varepsilon)\ln 2} + O(\theta^2).$$
(35)



Fig. 4. Comparison of sufficient conditions and necessary conditions as functions of *p*. The parameters are set as  $\varepsilon = 0.1$ ,  $\theta = 15$ dB,  $r_0 = 1$ , W = 0,  $\alpha = 4$  and  $\lambda = 0.05$ .

Therefore,  $\tilde{\zeta}_{c,\varepsilon}^{s} \log_2(1+\theta)$  and  $\tilde{\zeta}_{c,\varepsilon}^{n1} \log_2(1+\theta)$  approach 0 linearly with the same slope coefficient  $\frac{p}{\ln 2}$ , while  $\tilde{\zeta}_{c,\varepsilon}^{n2} \log_2(1+\theta)$  approaches 0 linearly with the slope coefficient  $\frac{p}{(1-\varepsilon)\ln 2}$ .

## B. Comparison of Sufficient and Necessary Conditions

In this subsection, we numerically compare the sufficient conditions and the necessary conditions derived in the previous sections. By the Monte Carlo simulation, we also plot the curves for the exact critical arrival rates, which are obtained by gradually increase the arrival rate until the condition for  $\varepsilon$ -stability of the network is not satisfied.

Fig. 4 shows the maximal arrival rates per the sufficient conditions and necessary conditions as functions of p. As  $p \rightarrow 0$ , all curves converge to 0, as explained in subsubsection VI-A1. As p increases, the curves for the non-closed-form sufficient condition (solid line with circle marks) and for the type I nonclosed-form necessary condition (solid line with square marks) first increase then decrease, because the success probability is limited by the small access probability for small p and by the large interference for large p.

Fig. 5 plots the maximal arrival rates per the sufficient conditions and necessary conditions as functions of  $\varepsilon$ . As  $\varepsilon \to 0$ , the curves for the sufficient conditions and the non-closed-form type II necessary condition approach 0, and other curves approach different constant values. Fig. 5 reveals that the curves do not depend strongly on  $\varepsilon$ . Since the gap between the curves for the sufficient conditions and that for the necessary conditions is not large, it can be inferred that the critical arrival rate for actual  $\varepsilon$ -stability region does not change much either when increasing  $\varepsilon$ . This observation indicates that a slight change in the arrival rate  $\zeta$  may greatly affect the fraction of unstable queues in the network.

Fig. 6 plots the maximal arrival rates per the sufficient conditions and necessary conditions as functions of  $\lambda$ . We observe that all curves except the one for the closed-form type II necessary condition converge to the same value. This is because as  $\lambda \rightarrow 0$ , the interference is negligible, and only



Fig. 5. Comparison of sufficient conditions and necessary conditions as functions of  $\varepsilon$ . The parameters are set as p = 0.5,  $\theta = 15$ dB,  $r_0 = 1$ , W = 0,  $\alpha = 4$  and  $\lambda = 0.05$ .



Fig. 6. Comparison of sufficient conditions and necessary conditions as functions of  $\lambda$ . The parameters are set as p = 0.5,  $\varepsilon = 0.1$ ,  $\theta = 15$ dB,  $r_0 = 1$ , W = 0 and  $\alpha = 4$ .

the noise affects the transmission for the dominant system and the simplified system; however, the curve for the closed-form type II necessary condition does not converge to the same value as  $\lambda \to 0$  because of the use of the Markov inequality in the derivation.

Fig. 7 plots the maximal arrival rates per the sufficient conditions and necessary conditions as functions of the distance between the transmitter and the receiver  $r_0$ . We observe that the type I necessary condition is better than the type II necessary condition for small  $r_0$  and worse for large  $r_0$ . It can be interpreted as that when the distance between the transmitter and receiver  $r_0$  is small, the power of the useful signal is large, and the interference from most of the transmitters is negligible compared to the useful signal. Therefore, considering only the nearest interferer, which is the case of the type I necessary condition, will result in a better approximation when  $r_0$  is small. On the other hand, when  $r_0$  is large, the effect of more interfering transmitters cannot be negligible; thus, the type II necessary condition will become better.

For the case where p and  $\theta$  can be optimized, i.e., the transmit probability p and the SINR threshold  $\theta$  are designable



Fig. 7. Comparison of sufficient conditions and necessary conditions as functions of  $r_0$ . The parameters are set as p = 0.5,  $\varepsilon = 0.1$ ,  $\theta = 15$ dB,  $\lambda = 0.05$ , W = 0 and  $\alpha = 4$ .



Fig. 8. Comparison of sufficient condition and necessary condition as a function of  $\lambda$  with optimal pair of  $(p, \theta)$ . The parameters are set as  $\varepsilon = 0.1$ ,  $r_0 = 1$ , W = 0 and  $\alpha = 4$ .

parameters that can be chosen to maximize the maximal arrival rate times  $\log_2(1+\theta)$ . To obtain realistic values, we choose  $\theta$ from [-20, 30] dB; then, Fig. 8 plots the maximal arrival rates times  $\log_2(1+\theta)$  in terms of sufficient conditions and necessary conditions as functions of  $\lambda$  when the optimal p and  $\theta$  are chosen.

As  $\varepsilon \to 0$ , the type II necessary condition is better than the type I necessary condition since the arrival rate can be positive to make the network strictly stable ( $\varepsilon = 0$ ) when only the nearest interferer is considered, which is not realistic in the original system. As  $p \to 0$ , a packet is dropped with high probability, and as  $\lambda \to 0$ , the interference caused by the interferers except the nearest one is negligible; thus, in these cases, the type I necessary condition is better. As  $\theta \to 0$  and  $p \to 1$ , the dropping of packets rarely happens, and if  $r_0$  is larger than the mean distance to the nearest interferer  $1/(2\sqrt{\lambda})$ , other interferers cannot be ignored; thus in these cases, the type II necessary condition will be better. We summarize the results in Table I, which lists some situations where it is preferable to use one of the two types of necessary conditions.

 TABLE I

 Some Situations to Use Type I or Type II Necessary Conditions

Case	Туре	Case	Туре
$\varepsilon \rightarrow 0$	Type II	$p \rightarrow 0$	Type I
$\lambda \to 0$	Type I	$\theta \to 0 \text{ and } p \to 1$	Type II
$\lambda > 1/(4r_0^2)$	Type II		

# VII. CONCLUSIONS

In this paper, we investigated the stable packet arrival rate region of a discrete-time slotted random access network with transmitters and receivers distributed as a static Poisson bipolar process. We introduced the notion of  $\varepsilon$ -stability, and obtained sufficient conditions and two types of necessary conditions for  $\varepsilon$ -stability. The asymptotic behaviors show that the obtained sufficient conditions and necessary conditions converge to the exact condition for  $\varepsilon$ -stability when some extreme system parameters are chosen. The numerical results reveal that the gap between the sufficient conditions and the necessary conditions is small when the access probability, the density of transmitters or the SINR threshold is small. The results also reveal that a slight change in the arrival rate may greatly affect the fraction of unstable queues in the network. Moreover, since we have two kinds of necessary conditions, we provide some guidance on whether to use the type I or type II necessary conditions for a specific network scenario.

# APPENDIX A Proof of Theorem 1

The success probability for the typical transmission conditioned on  $\Phi$  in the dominant system is denoted as  $\mathbb{P}^{x_0}(\mathcal{C}_{\Phi}) = p\mathbb{P}^{x_0}(SINR > \theta \mid \Phi)$ , which is evaluated as

$$\mathbb{P}^{x_{0}}(\mathcal{C}_{\Phi}) = p \mathbb{P}^{x_{0}} \left( h_{k,x_{0}} r_{0}^{-\alpha} > \theta \left( W + I_{k} \right) \mid \Phi \right)$$

$$\stackrel{(a)}{=} p \mathbb{E}^{x_{0}} \left[ \exp \left( -\theta r_{0}^{\alpha} \left( W + I_{k} \right) \right) \mid \Phi \right]$$

$$= p \mathbb{E}^{x_{0}} \left[ \exp \left( -\theta r_{0}^{\alpha} W - \sum_{x \in \Phi \setminus \{x_{0}\}} \theta r_{0}^{\alpha} h_{k,x} |x|^{-\alpha} \mathbf{1}(x \in \Phi_{k}) \right) \mid \Phi \right]$$

$$= \xi_{0} \prod_{x \in \Phi \setminus \{x_{0}\}} \left( p \mathbb{E}^{x_{0}} \left[ \exp \left( -\theta r_{0}^{\alpha} h_{k,x} |x|^{-\alpha} \right) \mid \Phi \right] + 1 - p \right)$$

$$\stackrel{(b)}{=} \xi_{0} \prod_{x \in \Phi \setminus \{x_{0}\}} \left( \frac{p}{1 + \theta r_{0}^{\alpha} |x|^{-\alpha}} + 1 - p \right). \tag{36}$$

where (*a*) and (*b*) follow because the fading coefficients  $\{h_{k,x}\}$  are i.i.d. exponential distributed random variables with unit mean. The moment generating function of  $Y \stackrel{\Delta}{=} \ln \left( \mathbb{P}^{x_0}(C_{\Phi}) \right)$  is

$$M_{Y}(s) = \mathbb{E}\left[e^{s\ln(\mathbb{P}^{x_{0}}(C_{\Phi}))}\right]$$
  
=  $(\xi_{0})^{s}\mathbb{E}\left[\prod_{x\in\Phi\setminus\{x_{0}\}}\left(\frac{p}{1+\theta r_{0}^{a}|x|^{-a}}+1-p\right)^{s}\right]$   
=  $(\xi_{0})^{s}\exp\left(-2\pi\lambda\int_{0}^{\infty}\left(1-\left(\frac{p}{1+\theta r_{0}^{a}r^{-a}}+1-p\right)^{s}\right)rdr\right)$   
=  $(\xi_{0})^{s}\exp\left(-sC_{\delta 2}F_{1}(1-s,\ 1-\delta;\ 2;\ p)\right).$  (37)

The cdf of *Y*, denoted by  $F_Y(y) = \mathbb{P}(Y \le y)$ , follows from the Gil-Pelaez Theorem [32] as

$$F_Y(y) = \frac{1}{2} - \frac{1}{\pi} \int_0^\infty \frac{\operatorname{Im}\{e^{-j\omega y} M_Y(j\omega)\}}{\omega} d\omega.$$
(38)

The probability that the transmitter  $x_0$  in the dominant system is unstable is given by the cdf of  $\mathbb{P}^{x_0}(\mathcal{C}_{\Phi})$ , which is

$$\mathbb{P}^{x_0}\left\{\mathbb{P}^{x_0}(\mathcal{C}_{\Phi}) \leq \tilde{\zeta}\right\} = \mathbb{P}^{x_0}\left\{Y \leq \ln \tilde{\zeta}\right\}$$
$$= \frac{1}{2} - \frac{1}{\pi} \int_0^\infty \frac{\operatorname{Im}\left\{e^{-j\omega \ln \tilde{\zeta}} M_Y(j\omega)\right\}}{\omega} d\omega.$$
(39)

The condition for the queue at the typical transmitter in the dominant system to be stable is  $\mathbb{P}^{x_0} \{\mathbb{P}^{x_0}(\mathcal{C}_{\Phi}) \leq \xi\} \leq \varepsilon$ . By combining (37) and (39), we obtain

$$\frac{1}{2} - \frac{1}{\pi} \int_0^\infty \frac{1}{\omega} \operatorname{Im} \{ (\xi_0)^{j\omega} \exp \left( -j\omega \ln \xi - j\omega C_{\delta 2} F_1 (1-j\omega, 1-\delta; 2; p) \right) \} d\omega \le \varepsilon.$$
(40)

Therefore, we get the results in the theorem.

# APPENDIX B Proof of Lemma 1

In the dominant simplified system, the transmitter  $x_0$  is active with probability p. The probability that the nearest interfering transmitter  $x_1$  is scheduled and also successful is

$$p_{1} = p^{2} \mathbb{P}\left\{\frac{h_{1}r_{0}^{-\alpha}}{h_{2}r_{s}^{-\alpha} + W} > \theta\right\} + p(1-p) \mathbb{P}\left\{\frac{h_{1}r_{0}^{-\alpha}}{W} > \theta\right\}$$
$$\stackrel{(a)}{=} \zeta_{0}\left(\frac{p}{1+\theta r_{0}^{\alpha}r_{s}^{-\alpha}} + 1 - p\right).$$
(41)

where  $h_1$  is the fading coefficient between the transmitter and the receiver of the nearest pair of transceiver, and  $h_2$  is the fading coefficient between the transmitter  $x_0$  and the receiver  $y_1$ . (a) follows because  $h_1$  and  $h_2$  are both exponentially distributed. In the following, we divide the proof into two cases, i.e.,  $\xi \ge p_1$  and  $\xi < p_1$ .

1) The Case Where  $\xi \ge p_1$ : When  $\xi \ge p_1$ , the queue at the nearest interfering transmitter  $x_1$  is unstable and will never be empty, thus  $x_1$  will cause interference to the typical transmission with probability p. Therefore, when  $\xi \ge p_1$  the probability that the transmitter  $x_0$  is scheduled and successful is

$$p_{0} = p^{2} \mathbb{P} \left\{ \frac{h_{3} r_{0}^{-\alpha}}{h_{4} r_{m}^{-\alpha} + W} > \theta \right\} + p(1-p) \mathbb{P} \left\{ \frac{h_{3} r_{0}^{-\alpha}}{W} > \theta \right\}$$
$$= \xi_{0} \left( \frac{p}{1 + \theta r_{0}^{\alpha} r_{m}^{-\alpha}} + 1 - p \right).$$
(42)

where  $h_3$  is the fading coefficient between the transmitter  $x_0$  and the receiver  $y_0$ , and  $h_4$  is the fading coefficient between the nearest interfering transmitter  $x_1$  and the receiver  $y_0$ .

Authorized licensed use limited to: Huazhong University of Science and Technology. Downloaded on March 25,2020 at 09:18:21 UTC from IEEE Xplore. Restrictions apply.

If  $r_s > r_m$ , by comparing (41) with (42), we have  $\xi > p_1 > p_0$ , which implies that the queue at the transmitter  $x_0$  is unstable if  $\xi \ge p_1$ . This can be explained intuitively by the concept of "stability rank" [9], which states that if a queue is unstable, the queues with higher rank than the said queue are unstable as well. The interference from the transmitter  $x_0$  to the receiver  $y_1$  is less than that from the nearest interfering transmitter  $x_1$  to the receiver  $y_0$  when  $r_s > r_m$ , which means that the queue at  $x_0$  has higher rank than the queue at  $x_1$ ; Thus, when the queue at the nearest interference  $x_1$  is unstable, the queue at the transmitter  $x_0$  is also unstable.

If  $r_s \leq r_m$ , the queue at  $x_1$  has higher rank than the queue at  $x_0$ . By comparing (41) with (42), we have  $p_0 \geq p_1$ , which implies that the queue at the transmitter  $x_0$  is stable for  $p_0 \geq \zeta \geq p_1$  and unstable for  $\zeta > p_0$ .

2) The Case Where  $\xi < p_1$ : When  $\xi < p_1$ , the queue of the nearest interfering transmitter  $x_1$  is empty with probability  $1 - \xi/p_1$  and is nonempty with probability  $\xi/p_1$ . Therefore, when  $\xi < p_1$  the probability that the transmitter  $x_0$  is scheduled by random access and successful is

$$p_{0}' = p^{2} \frac{\xi}{p_{1}} \mathbb{P}\left\{\frac{h_{3}r_{0}^{-\alpha}}{h_{4}r_{m}^{-\alpha} + W} > \theta\right\}$$
$$+ \left(p(1-p)\frac{\xi}{p_{1}} + p\left(1-\frac{\xi}{p_{1}}\right)\right) \mathbb{P}\left\{\frac{h_{3}r_{0}^{-\alpha}}{W} > \theta\right\}$$
$$= \xi_{0}\left(\frac{p\xi}{p_{1}}\frac{1}{1+\theta r_{0}^{\alpha}r_{m}^{-\alpha}} + 1 - \frac{p\xi}{p_{1}}\right).$$
(43)

To make the queue at the transmitter  $x_0$  stable, the arrival rate should satisfy  $\xi \leq p'_0$ , i.e.,

$$\begin{split} \xi &\leq \frac{pp_1}{p_1 \exp\left(W\theta r_0^{\alpha}\right) + p^2 - p^2 \frac{1}{1 + \theta r_0^{\alpha} r_m^{-\alpha}}} \\ &= p_1 \left(\frac{p}{1 + \theta r_0^{\alpha} r_s^{-\alpha}} - \frac{p}{1 + \theta r_0^{\alpha} r_m^{-\alpha}} + 1\right)^{-1} \\ &= \xi_0 \left(\frac{p}{1 + \theta r_0^{\alpha} r_s^{-\alpha}} + 1 - p\right) \\ &\times \left(\frac{p}{1 + \theta r_0^{\alpha} r_s^{-\alpha}} - \frac{p}{1 + \theta r_0^{\alpha} r_m^{-\alpha}} + 1\right)^{-1}. \end{split}$$
(44)

If  $r_s > r_m$ , it can be verified that the right side of the above inequality is less than  $p_1$ . Therefore, if  $\xi < p_1$ , the queue at the transmitter  $x_0$  in the dominant simplified system will be stable only when the inequality (44) is satisfied.

If  $r_s \leq r_m$ , the right side of the above inequality is larger than  $p_1$ . Therefore, for the case  $\xi < p_1$ , the queue at the transmitter  $x_0$  in the dominant simplified system will be stable.

Combining the cases  $\xi \ge p_1$  and  $\xi < p_1$ , the queue at  $x_0$  in the dominant simplified system is stable if and

only if

$$\xi \leq \begin{cases} \xi_{0} \left( \frac{p}{1 + \theta r_{0}^{\alpha} r_{s}^{-\alpha}} + 1 - p \right) \\ \cdot \left( \frac{p}{1 + \theta r_{0}^{\alpha} r_{s}^{-\alpha}} - \frac{p}{1 + \theta r_{0}^{\alpha} r_{m}^{-\alpha}} + 1 \right)^{-1} & \text{if } r_{s} > r_{m} \\ \xi_{0} \left( \frac{p}{1 + \theta r_{0}^{\alpha} r_{m}^{-\alpha}} + 1 - p \right) & \text{if } r_{s} \leq r_{m} \end{cases}$$

$$(45)$$

cessary condition

Thus, (45) also gives the sufficient and necessary condition for the queue at the transmitter  $x_0$  to be stable in the original simplified system.

## APPENDIX C

#### **PROOF OF THEOREM 2**

According to Lemma 1, if  $r_s > r_m$ , from (45) we have

$$\begin{split} \xi &\leq \xi_0 \left( \frac{p}{1 + \theta r_0^{\alpha} r_s^{-\alpha}} + 1 - p \right) \\ &\times \left( \frac{p}{1 + \theta r_0^{\alpha} r_s^{-\alpha}} - \frac{p}{1 + \theta r_0^{\alpha} r_m^{-\alpha}} + 1 \right)^{-1} \\ &\leq \xi_0 \left( \frac{p}{1 + \theta r_0^{\alpha} r_s^{-\alpha}} + 1 - p \right). \end{split}$$
(46)

Since Lemma 1 gives a sufficient and necessary condition for the transmitter  $x_0$  to be stable in the simplified system when  $\varphi$ ,  $\psi$ ,  $r_m$  are given, comparing (45) and (46), we obtain a necessary condition as

$$\xi \leq \xi_0 \left( \frac{p}{1 + \theta r_0^{\alpha} \left( \max\{r_{\rm m}, r_{\rm s}\} \right)^{-\alpha}} + 1 - p \right). \tag{47}$$

According to (3) and Lemma 1, when  $\varphi$ ,  $\psi$ ,  $r_m$  are random variables, a necessary condition for the simplified system to be stable is

$$\varepsilon \geq \mathbb{P}\left\{ \xi \geq \xi_0 \left( \frac{p}{1 + \theta r_0^{\alpha} \left( \max\{r_{\mathrm{m}}, r_{\mathrm{s}} \} \right)^{-\alpha}} + 1 - p \right) \right\}$$
$$= \mathbb{P}\left\{ \frac{1}{r_0} \max\{r_{\mathrm{m}}, r_{\mathrm{s}} \} \leq \left( \theta \frac{\xi + p\xi_0 - \xi_0}{\xi_0 - \xi} \right)^{1/\alpha} \right\}.$$
(48)

Let  $Z = \frac{1}{r_0} \max\{r_m, r_s\}$ , and denote the cdf of Z as  $F_Z(z)$ . (48) can be written as

$$\varepsilon \ge F_Z\left(\frac{1}{r_0}\max\{r_{\rm m}, r_{\rm s}\} \le \left(\theta \frac{\xi + p\xi_0 - \xi_0}{\xi_0 - \xi}\right)^{1/\alpha}\right), \quad (49)$$

which is equivalent to

Ż

$$\xi \leq \xi_0 \left( 1 - \frac{\theta p}{\theta + \left(F_Z^{-1}(\varepsilon)\right)^{\alpha}} \right).$$
 (50)

Therefore, we obtain the necessary condition in Theorem 2.

# Appendix D

# PROOF OF COROLLARY 2

According to (3) and Lemma 2,  $r_{\rm m}$  is a random variable in the original system. Thus, a necessary condition for the original system to be  $\varepsilon$ -stable is

Since  $f(x) = \frac{x}{1+x}$  is an increasing function, we obtain a necessary condition for the original system to be  $\varepsilon$ -stable as

$$\varepsilon > \mathbb{P}\left\{ \xi \ge \frac{\left( (r_{\rm m} + 2r_0)^{\alpha} + \theta r_0^{\alpha} \right)^2 \xi_0}{\left( (r_{\rm m} + 2r_0)^{\alpha} + \theta r_0^{\alpha} \right)^2 + p^2 \theta^2 r_0^{2\alpha}} \right\} = \mathbb{P}\left\{ (\xi_0 - \xi) \left( (r_{\rm m} + 2r_0)^{\alpha} + \theta r_0^{\alpha} \right)^2 \le \xi p^2 \theta^2 r_0^{2\alpha} \right\}.$$
 (52)

Since the inequality  $\xi_0 - \xi > 0$  is satisfied from (4), we have

$$\varepsilon > \mathbb{P}\left\{r_{\mathrm{m}} \leq \underbrace{\left(\sqrt{\frac{\xi p^{2}\theta^{2}r_{0}^{2\alpha}}{\xi_{0}-\xi}} - \theta r_{0}^{\alpha}\right)^{1/\alpha} - 2r_{0}}_{A}\right\}.$$
 (53)

When  $A \leq 0$ , the probability at the right side of the inequality is zero; thus the above inequality (53) always holds. When A > 0, using the probability distribution of  $r_m$  given by (17), we have

$$\varepsilon > 1 - \exp\left(-\pi \lambda A^2\right),$$
 (54)

and thus

$$0 < A < \sqrt{-\frac{\ln(1-\varepsilon)}{\pi\,\lambda}}.\tag{55}$$

Combining the cases of  $A \leq 0$  and A > 0, we have

$$\left(\sqrt{\frac{\xi p^2 \theta^2 r_0^{2\alpha}}{\xi_0 - \xi}} - \theta r_0^{\alpha}\right)^{1/\alpha} - 2r_0 < \sqrt{-\frac{\ln(1 - \varepsilon)}{\pi \lambda}}.$$
 (56)

Solving the above inequality, we get the result in the corollary.

# APPENDIX E Proof of Theorem 3

By introducing the modified favorable system, an interfering transmitter is active with probability  $\xi p$ . Similar to the derivation of (36), we get

$$\mathbb{P}^{x_{0}}(\mathcal{C}_{\Phi})$$

$$= \xi_{0} \prod_{x \in \Phi \setminus \{x_{0}\}} \left( \xi p \mathbb{E}^{x_{0}} \Big[ \exp\left(-\theta r_{0}^{\alpha} h_{k,x} |x|^{-\alpha}\right) | \Phi \Big] + 1 - \xi p \right)$$

$$= \xi_{0} \prod_{x \in \Phi \setminus \{x_{0}\}} \left( \frac{\xi p}{1 + \theta r_{0}^{\alpha} |x|^{-\alpha}} + 1 - \xi p \right).$$
(57)

Letting  $Y \stackrel{\Delta}{=} \ln (\mathbb{P}^{x_0}(\mathcal{C}_{\Phi}))$ , the moment generating function of *Y* is

$$M_Y(s) = (\xi_0)^s \exp\left(-\xi s C_{\delta 2} F_1(1-s, 1-\delta; 2; \xi p)\right).$$
(58)

The cdf of Y can be derived as follows by applying the Gil-Pelaez Theorem given by (38).

$$F_Y(y) = \frac{1}{2} - \frac{1}{\pi} \int_0^\infty \frac{\operatorname{Im}\{e^{-j\omega y} M_Y(j\omega)\}}{\omega} d\omega.$$
 (59)

The probability that the queue at the typical transmitter in the modified system is unstable is

$$\mathbb{P}^{x_0} \left\{ \mathbb{P}^{x_0}(\mathcal{C}_{\Phi}) \leq \xi \right\}$$
  
=  $\mathbb{P}^{x_0} \left\{ \ln \left( \mathbb{P}^{x_0}(\mathcal{C}_{\Phi}) \right) \leq \ln \xi \right\}$   
=  $\frac{1}{2} - \frac{1}{\pi} \int_0^\infty \frac{\operatorname{Im}\{e^{-j\omega \ln \xi} M_Y(j\omega)\}}{\omega} d\omega.$  (60)

The condition for the queue at the typical transmitter in the modified system to be stable is  $\mathbb{P}^{x_0} \{\mathbb{P}^{x_0}(\mathcal{C}_{\Phi}) \leq \xi\} \leq \varepsilon$ . By combining (58) and (60), we get the condition for the queue at the typical transmitter in the modified system to be stable as

$$\frac{1}{2} - \frac{1}{\pi} \int_0^\infty \frac{1}{\omega} \operatorname{Im} \{ (\xi_0)^{j\omega} \exp\left(-j\omega \ln \xi - j\omega \xi C_{\delta 2} F_1 (1-j\omega, 1-\delta; 2; \xi p) \right) \} d\omega \le \varepsilon.$$
(61)

Therefore, we get the necessary condition for the original system to be  $\varepsilon$ -stable.

#### APPENDIX F

#### PROOF OF COROLLARY 3

For all t > 0, by applying Markov inequality, we obtain

$$\mathbb{P}^{x_{0}} \left\{ \mathbb{P}^{x_{0}}(\mathcal{C}_{\Phi}) < \xi \right\} \\
= \mathbb{P}^{x_{0}} \left\{ (\mathbb{P}^{x_{0}}(\mathcal{C}_{\Phi}))^{t} < \xi^{t} \right\} \\
> 1 - \xi^{-t} \mathbb{E} \left[ \left( \mathbb{P}^{x_{0}}(\mathcal{C}_{\Phi}) \right)^{t} \right] \\
= 1 - (\xi_{0})^{t} \xi^{-t} \mathbb{E} \left[ \prod_{x \in \Phi \setminus \{x_{0}\}} \left( \frac{\xi p}{1 + \theta r_{0}^{\alpha} |x|^{-\alpha}} + 1 - \xi p \right)^{t} \right] \\
= 1 - (\xi_{0})^{t} \xi^{-t} \exp \left( - \xi t C_{\delta 2} F_{1}(1 - t, 1 - \delta; 2; \xi p) \right).$$
(62)

Solving the following inequality, we get a type II necessary condition given by (23):

$$1 - (\xi_0)^t \xi^{-t} \exp\left(-\xi t C_{\delta 2} F_1(1-t, 1-\delta; 2; \xi p)\right) \le \varepsilon.$$
  
When  $t = 1$ , we obtain

$$\mathbb{P}^{x_0}\left\{\mathbb{P}^{x_0}(\mathcal{C}_{\Phi}) < \zeta\right\} > 1 - \zeta_0 \zeta^{-1} \exp(-\zeta C_{\delta}).$$
(63)

Let  $\mathcal{W}(z)$  be the main branch of Lambert W function, defined by  $z = \mathcal{W}(z)e^{\mathcal{W}(z)}$  for any complex number z. Solving the inequality

$$1 - \xi_0 \xi^{-1} \exp(-\xi C_\delta) \le \varepsilon, \tag{64}$$

we get a closed-form type II necessary condition in the (58) corollary.

#### REFERENCES

- [1] Y. Zhong, W. Zhang, and M. Haenggi, "Stability analysis of static Poisson networks," in *Proc. IEEE Int. Symp. Inf. Theory (ISIT)*, Hong Kong, Jun. 2015, pp. 2812–2816. [Online]. Available: http://www.nd.edu/~mhaenggi/pubs/isit15c.pdf
- [2] M. Haenggi, J. G. Andrews, F. Baccelli, O. Dousse, and M. Franceschetti, "Stochastic geometry and random graphs for the analysis and design of wireless networks," *IEEE J. Sel. Areas Commun.*, vol. 27, no. 7, pp. 1029–1046, Sep. 2009.
- [3] M. Haenggi, Stochastic Geometry for Wireless Networks. Cambridge, U.K.: Cambridge Univ. Press, 2012.
- [4] F. Baccelli and B. Błaszczyszyn, Stochastic Geometry and Wireless Networks, vol. 1. Norwell, MA, USA: NOW Publishers, 2009.
- [5] R. R. Rao and A. Ephremides, "On the stability of interacting queues in a multiple-access system," *IEEE Trans. Inf. Theory*, vol. 34, no. 5, pp. 918–930, Sep. 1988.
- [6] B. S. Tsybakov and V. A. Mikhajlov, "Ergodicity of a slotted ALOHA system," *Problemy Peredachi Informatsii*, vol. 15, no. 4, pp. 73–87, 1979.
- [7] W. Szpankowski, "Stability conditions for some distributed systems: Buffered random access systems," *Adv. Appl. Probab.*, vol. 26, no. 2, pp. 498–515, Jun. 1994.
- [8] V. Anantharam, "The stability region of the finite-user slotted ALOHA protocol," *IEEE Trans. Inf. Theory*, vol. 37, no. 3, pp. 535–540, May 1991.
- [9] W. Luo and A. Ephremides, "Stability of N interacting queues in random-access systems," *IEEE Trans. Inf. Theory*, vol. 45, no. 5, pp. 1579–1587, Jul. 1999.
- [10] F. Baccelli, B. Błaszczyszyn, and P. Muhlethaler, "Stochastic analysis of spatial and opportunistic Aloha," *IEEE J. Sel. Areas Commun.*, vol. 27, no. 7, pp. 1105–1119, Sep. 2009.
- [11] D. P. Bertsekas, R. G. Gallager, and P. Humblet, *Data Networks*, vol. 2. Englewood Cliffs, NJ, USA: Prentice-Hall, 1992.
- [12] W. Szpankowski, "Stability conditions for multidimensional queueing systems with computer applications," *Oper. Res.*, vol. 36, no. 6, pp. 944–957, 1988.
- [13] W. Szpankowski, "Bounds for queue lengths in a contention packet broadcast system," *IEEE Trans. Commun.*, vol. 34, no. 11, pp. 1132–1140, Nov. 1986.
- [14] L. Tassiulas and A. Ephremides, "Stability properties of constrained queueing systems and scheduling policies for maximum throughput in multihop radio networks," *IEEE Trans. Autom. Control*, vol. 37, no. 12, pp. 1936–1948, Dec. 1992.
- [15] M. J. Neely and E. Modiano, "Capacity and delay tradeoffs for ad hoc mobile networks," *IEEE Trans. Inf. Theory*, vol. 51, no. 6, pp. 1917–1937, Jun. 2005.
- [16] M. Haenggi and R. Smarandache, "Diversity polynomials for the analysis of temporal correlations in wireless networks," *IEEE Trans. Wireless Commun.*, vol. 12, no. 11, pp. 5940–5951, Nov. 2013.
- [17] N. Pappas, M. Kountouris, and A. Ephremides, "The stability region of the two-user interference channel," in *Proc. IEEE Inf. Theory Workshop (ITW)*, Sep. 2013, pp. 1–5.
- [18] I. E. Telatar and R. G. Gallager, "Combining queueing theory with information theory for multiaccess," *IEEE J. Sel. Areas Commun.*, vol. 13, no. 6, pp. 963–969, Aug. 1995.
- [19] N. Xie, J. M. Walsh, and S. Weber. (2014). "Properties of an Aloha-like stability region." [Online]. Available: http://arxiv.org/abs/1408.3469
- [20] S. B. Lowen and M. C. Teich, "Power-law shot noise," *IEEE Trans. Inf. Theory*, vol. 36, no. 6, pp. 1302–1318, Nov. 1990.
- [21] M. Haenggi, "The meta distribution of the SIR in Poisson bipolar and cellular networks," *IEEE Trans. Wireless Commun.*, vol. 15, no. 4, pp. 2577–2589, Apr. 2016.
- [22] S. P. Weber, X. Yang, J. G. Andrews, and G. de Veciana, "Transmission capacity of wireless ad hoc networks with outage constraints," *IEEE Trans. Inf. Theory*, vol. 51, no. 12, pp. 4091–4102, Dec. 2005.
- [23] M. Haenggi and R. K. Ganti, *Interference in Large Wireless Networks*, vol. 3. Norwell, MA, USA: NOW Publishers, 2009.
- [24] R. K. Ganti and M. Haenggi, "Spatial and temporal correlation of the interference in ALOHA ad hoc networks," *IEEE Commun. Lett.*, vol. 13, no. 9, pp. 631–633, Sep. 2009.
- [25] M. Haenggi, "Diversity loss due to interference correlation," *IEEE Commun. Lett.*, vol. 16, no. 10, pp. 1600–1603, Oct. 2012.
  [26] F. Baccelli and B. Błaszczyszyn, "A new phase transitions for local
- [26] F. Baccelli and B. Błaszczyszyn, "A new phase transitions for local delays in MANETs," in *Proc. IEEE INFOCOM*, Mar. 2010, pp. 1–9.
- [27] M. Haenggi, "The local delay in Poisson networks," *IEEE Trans. Inf. Theory*, vol. 59, no. 3, pp. 1788–1802, Mar. 2013.

- [28] Y. Zhong, W. Zhang, and M. Haenggi, "Managing interference correlation through random medium access," *IEEE Trans. Wireless Commun.*, vol. 13, no. 2, pp. 928–941, Feb. 2014.
- [29] Z. Gong and M. Haenggi, "The local delay in mobile Poisson networks," *IEEE Trans. Wireless Commun.*, vol. 12, no. 9, pp. 4766–4777, Sep. 2013.
- [30] K. Stamatiou and M. Haenggi, "Random-access Poisson networks: Stability and delay," *IEEE Commun. Lett.*, vol. 14, no. 11, pp. 1035–1037, Nov. 2010.
- [31] R. M. Loynes, "The stability of a queue with non-independent interarrival and service times," *Math. Proc. Cambridge Philos. Soc.*, vol. 58, no. 3, pp. 497–520, 1962.
- [32] J. G. Wendel *et al.*, "The non-absolute convergence of Gil–Pelaez' inversion integral," *Ann. Math. Statist.*, vol. 32, no. 1, pp. 338–339, Apr. 1961.



Yi Zhong (S'11–M'15) received the B.S. and Ph.D. degrees in electronics engineering from the University of Science and Technology of China, Hefei, China, in 2010 and 2015, respectively.

He was a Visiting Student with Prof. M. Haenggi's group, University of Notre Dame, Notre Dame, IN, USA, in 2012. In 2013, he was an Intern with Qualcomm Inc., Corporate Research and Development, Beijing. He is currently a Post-Doctoral Research Fellow with the WNDS Group, Singapore University of Technology and Design,

Singapore, led by Prof. T. Q. S. Quek. His research interests include heterogeneous and femtocell-overlaid cellular networks, wireless ad hoc networks, stochastic geometry, and point process theory.



Martin Haenggi (S'95–M'99–SM'04–F'14) received the Dipl.-Ing. (M.Sc.) and Dr.Sc.Techn. (Ph.D.) degrees in electrical engineering from the Swiss Federal Institute of Technology (ETH), Zurich, in 1995 and 1999, respectively. From 2007 to 2008, he was a Visiting Professor with the University of California at San Diego. From 2014 to 2015 he was an Invited Professor with EPFL, Switzerland. He is currently a Freimann Professor of Electrical Engineering and a Concurrent Professor of Applied and Computational

Mathematics and Statistics with the University of Notre Dame, IN, USA.

He has co-authored the monograph entitled *Interference in Large Wireless Networks* (NOW Publishers, 2009) and authored the textbook entitled *Stochastic Geometry for Wireless Networks* (Cambridge University Press, 2012). His scientific interests include networking and wireless communications, with an emphasis on cellular, amorphous, ad hoc, cognitive, and vehicular networks.

He served as an Associate Editor of the Journal of Ad Hoc Networks (Elsevier) from 2005 to 2008, the IEEE TRANSACTIONS ON MOBILE COMPUTING (TMC) from 2008 to 2011, and the ACM Transactions on Sensor Networks from 2009 to 2011, and as a Guest Editor of the IEEE JOURNAL ON SELECTED AREAS IN COMMUNICATIONS from 2008 to 2009 and the IEEE TRANSACTIONS ON VEHICULAR TECHNOLOGY from 2012 to 2013. He also served as a Steering Committee Member of the TMC from 2011 to 2013, a Distinguished Lecturer of the IEEE Circuits and Systems Society from 2005 to 2006, the TPC Co-Chair of the Communication Theory Symposium of the 2012 IEEE International Conference on Communications and the 2014 International Conference on Wireless Communications and Signal Processing, a General Co-Chair of the 2009 International Workshop on Spatial Stochastic Models for Wireless Networks and the 2012 DIMACS Workshop on Connectivity and Resilience of Large-Scale Networks, and a Keynote Speaker of several international conferences and workshops. In 2017, he will be the Editor-in-Chief of the IEEE TRANSACTIONS ON WIRELESS COMMUNICATIONS.

For both his M.Sc. and Ph.D. theses, he received the ETH Medal. He also received the CAREER Award from the U.S. National Science Foundation in 2005 and the 2010 IEEE Communications Society Best Tutorial Paper Award.



Tony Q. S. Quek (S'98–M'08–SM'12) received the B.E. and M.E. degrees in electrical and electronics engineering from the Tokyo Institute of Technology, Tokyo, Japan, and the Ph.D. degree in electrical engineering and computer Science from the Massachusetts Institute of Technology. Currently, he is an Assistant Professor with the Information Systems Technology and Design Pillar, Singapore University of Technology and Design (SUTD), Singapore. He is also a Scientist with the Institute for Infocomm Research. His main research interests are the appli-

cation of mathematical, optimization, and statistical theories to communication, networking, signal processing, and resource allocation problems. His specific current research topics include heterogeneous networks, green communications, smart grid, wireless security, Internet-of-Things, big data processing, and cognitive radio.

Dr. Quek has been actively involved in organizing and chairing sessions, and has served as a member of the Technical Program Committee as well as the Symposium Chair in a number of international conferences. He is an Editor of the IEEE TRANSACTIONS ON COMMUNICATIONS and an Executive Editorial Committee Member of the IEEE TRANSACTIONS ON WIRELESS COMMUNICATIONS. He was an Editor of the IEEE WIRELESS COMMUNICATIONS LETTERS and a Guest Editor of the IEEE Signal Processing Magazine (Special Issue on Signal Processing for the 5G Revolution) in 2014, and the IEEE Wireless Communications Magazine (Special Issue on Heterogeneous Cloud Radio Access Networks) in 2015. He has co-authored the books entitled Small Cell Networks: Deployment, PHY Techniques, and Resource Allocation (Cambridge University Press, 2013) and Cloud Radio Access Networks: Principles, Technologies, and Applications (Cambridge University Press).

Dr. Quek received the 2008 Philip Yeo Prize for Outstanding Achievement in Research, the IEEE Globecom 2010 Best Paper Award, the 2012 IEEE William R. Bennett Prize, the IEEE SPAWC 2013 Best Student Paper Award, the IEEE WCSP 2014 Best Paper Award, the IEEE PES General Meeting 2015 Best Paper, and the 2015 SUTD Outstanding Education Awards—Excellence in Research.



Wenyi Zhang (S'00–M'07–SM'11) received the bachelor's degree in automation from Tsinghua University, Beijing, China, in 2001, and the master's and Ph.D. degrees in electrical engineering from the University of Notre Dame, Notre Dame, IN, USA, in 2003 and 2006, respectively. He was with the Communication Science Institute, University of Southern California, Los Angeles, CA, USA, as a Post-Doctoral Research Associate, and with Qualcomm Inc., Corporate Research and Development. He is a Faculty Member with the Department

of Electronic Engineering and Information Science, University of Science and Technology of China, Hefei, China. His research interests include wireless communications and networking, information theory, and statistical signal processing.